Author's response to reviews

Title: Henoch-Schonlein Purpura in an Elderly Man Presenting as Rectal Bleeding and IgA Mesangioproliferative Glomerulonephritis: A case report

Authors:

Wisit Cheungpasitporn (strong_wind@hotmail.com)
Teeranun Jirajariyavej (teeranun.jirajariyavej@bassett.org)
Charles B Howarth (charles.howarth@bassett.org)
Raquel M Rosen (raquel.rosen@bassett.org)

Version: 3 Date: 7 May 2011

Author's response to reviews: see over
Dear referees and reviewers of Journal of Medical Case Reports,

Thank you for your email and very useful reviewer’s reports. We have revised and resubmitted our manuscript to the journal style.

We reviewed the reviewer’s reports and subsequently revised the manuscript as following:

1. We revised the abstract part to be more concise for the reader.
2. We mentioned history of precipitating factors in this age group, such as ingestion of non steroidal anti-inflammatory drugs, angiotensin-converting enzyme inhibitors, use of prior antibiotics, history of food allergies, vaccinations or insect bites in the case presentation as reviewer’s suggestion.
3. We raised the aspect of drug induced vasculitis in the manuscript as reviewer’s suggestion.

Thank you very much,
Look forward to hearing from you,
Best regards,

Wisit Cheungpasitporn, M.D.
Department of Internal Medicine, Bassett Medical Center, Cooperstown, New York 13326, USA.